Author's response to reviews

Title: Did a quality improvement collaborative make stroke care better? A cluster randomised trial

Authors:

Maxine Power (maxine.power@nhs.net)
Pippa J Tyrrell (pippa.tyrrell@manchester.ac.uk)
Anthony G Rudd (anthony.rudd@kcl.ac.uk)
Mary P Tully (mary.tully@manchester.ac.uk)
David Dalton (david.dalton@srf.t.nhs.uk)
Martin Marshall (martin.marshall@islondon.org)
Ian Chappell (ian.chappell@nhs.net)
Delphine Corgié (delphinecorgie@nhs.net)
Don Goldmann (dgolddmann@ihi.org)
Dale Webb (dale.webb@stroke.org.uk)
Mary Dixon-Woods (md11@le.ac.uk)
Gareth Parry (gparry@ihi.org)

Version: 2 Date: 23 September 2013

Author's response to reviews:

In response to a request, we have included a title page as p.1.